Title: SNP-SNP Interactions in Breast Cancer Susceptibility

Authors:

Venus U Onay (onay@mshri.on.ca)
Laurent Briollais (laurent@mshri.on.ca)
Julia A Knight (knight@mshri.on.ca)
Ellen Shi (Ellen.Shi@cancercare.on.ca)
Yuanyuan Wang (y32wang@uwaterloo.ca)
Sean Wells (sean@mshri.on.ca)
Hong Li (li@mshri.on.ca)
Isaac Rajendram (idr1978@hotmail.com)
Irene L Andrulis (andrulis@mshri.on.ca)
Hilmi Ozcelik (ozcelik@mshri.on.ca)

Version: 5  Date: 13 April 2006

Author's response to reviews: see over
12 April 2006

Our Ref: CT/mm/apr 06.J024

Dr Ummiye Venus Onay
Mount Sinai Hospital
onay@mshri.on.ca

Dear Dr Onay

*MUTATION RESEARCH, Vol 544, 2003, pp 9-41, Mitrunen et al, “Molecular epidemiology...” Figure 3 only*

Proposed Use: For publication in BMC Cancer.

As per your letter dated 11 April 2006, we hereby grant you permission to reproduce the aforementioned material in **print and electronic format** at no charge subject to the following conditions:

1. If any part of the material to be used (for example, figures) has appeared in our publication with credit or acknowledgement to another source, permission must also be sought from that source. If such permission is not obtained then that material may not be included in your publication/copies.

2. Suitable acknowledgement to the source must be made, either as a footnote or in a reference list at the end of your publication, as follows:

   “Reprinted from Publication title, Vol number, Author(s), Title of article, Pages No., Copyright (Year), with permission from Elsevier”.

3. This permission is granted for non-exclusive world rights in all languages.

4. Reproduction of this material is granted for the purpose for which permission is hereby given, and includes use in any future editions.

Yours sincerely

Clare Truter
Deputy Rights Manager, S&T

*Your future requests will be handled more quickly if you complete the online form at [www.elsevier.com/permissions]*